helloUML Top-Level Design

Lisa Cozzens (ljc)

February25, 2000

1. High-level component diagrams

1.1. GUI subsystem

Layout Wizard, Design Wizard: Both of these components emplo ysome AI magic that

I’m not reall yclear on at this point. The yneed to know about the graph because the yneed

to take into account all the objects and connections to do their magic. The La yout Wizard

should layout all the objects in some sort of visuall yappealing manner. For example, it

might place all subclasses directlyunder their superclass. The Design Wizard helps the

user b ywatching what the ydo and making suggestions about design, sort of like (dare I

sayit) the stupid paperclip in Microsoft Word (but much less anno ying).

Loader, Saver: These components are responsible for loading (and parsing) files and

saving files. The loader needs to know about the graph so it can add objects defined in the

file to the graph. The saver needs to know about the graph so it can save the objects in the

graph to the file.

Code generator: This component is responsible for generating code based on the objects

and connections that the user has drawn. Consequentl y, it needs to know about the graph

so it can get access to the objects and connections. There are subclasses of the code

generator to handle different programming languages (Java, C++�.)

Undo event holder: The undo event holder hangs onto undo events in chronological

order. It provides methods to get the most recent event so it can be undone.

Undo event: The undo event knows about the state of the s ystem prior to an event, so it

can restore the s ystem to that state. For example, suppose the user changes the name of an

object. An undo event would be generated, and it would have a cop yof the object before

the name change. If the user wanted to undo that event, this copywould be moved back

into the graph. This creates a c yclic dependenc ybetween Graphical Objects and undo

events, but I don’t see how it can be avoided. The undo event class should be small and

simple enough so that this isn’t too much of a problem.

2. External dependencies

This project does not have an ydependencies on external programs.

3. Grouporganization

The group should be organized so as to fill the following roles:

· Project leader: responsible for general organization (e.g. Person X should work on

Part Y), reviewing group members’ work and suggesting improvements if necessar y,

keeping things on schedule and figuring out what to do if the schedule slips, resolving

anyproblems that arise within the team, etc.

· Design master: responsible for the overall design of the program. This person must

approve all proposed additions and changes to the list of required features, keeping in

mind the "conceptual integrit y" of the program. Although all group members can and

should contribute to the design, this person has final sa y.

· Lead programmer and co-pilot: the person most responsible for the actual

implementation of the program. How does this differ from the design master? First of

all, the chief programmer will be responsible for more programming. Secondly, the

design master is more concerned with high-level questions (e.g. will we need a

database? ) while the chief programmer can focus on more nitt y-gritt yquestions

(which database management software will we use? ) The co-pilot helps the chief

programmer make these decisions, along with the rest of the team, but the chief

programmer’s sa yis final.

· Editor: responsible for the program’s documentation. As with design, all team

members can and should contribute to the documentation, particularl yfor the parts of

the program that the yare directl yinvolved with. The editor’s job is to delegate

documentation to different team members, to give feedback to team members on the

documentation that theywrote, then to collate the documentation into a single

coherent and consistent document.

· Tester: responsible for overseeing testing of the integrated s ystem. Each team

member should test his/her individual components, but the tester has primary responsibilityfor finding bugs that arise after integration. The tester is also

responsible for coordinating user testing.

· User interface manager: the equivalent of the design master for the user interface,

this person is responsible for ensuring that the user interface is consistent and eas yto

use. S/he mayset guidelines, e.g. sa ythat ever yerror message should be displa yed in

a small dialog box that includes the error number and a clear description of what the

error means, along with an OK button.

· Toolsmith: responsible for learning about and setting up tools that maybe of use to

the programmers. For example, this person should select the configuration

management software that the group will use (with input from the rest of the group),

set it up, and help the other group members learn how to use it.

All team members will be involved in programming to some extent, but the chief

programmer and co-pilot will take on a larger share of the programming than other team

members .

4. Taskbreakdow n

· GUI subsystem: Two people (one of whom should be the user interface manager)

will be assigned to this section. One person will work on implementing the UML

playground and the various tools, while the other will work on the actual GU I (e.g.

this button should look like this and go in this place)

· Graph tools subsystem: Three people will be assigned to this section. One person

will work on loading and saving. One person will work on code generation. One

person will work on the La yout Wizard and/or Design Wizard.

· Graph subsystem: The remaining three people will work on this section. The lead

programmer should be one of these people. One person will work on the rule checker.

The other two people will work on the graph itself and undo functionality.

5. Schedule

February 25: Top-level design proposals

March 3: Discussion of top-level design proposals

March 10: Finalization of top-level design

March 15: Public interface proposals

March 20: Discussion of public interface proposals

March 24: Finalization of public interfaces

April 5: Detailed designs of components finalized, begin coding.

April 10: Initial integration, with much of the functionalit ystill stubbed out

April 17: Integration, with much of the functionalityand the entire user interface

implemented

April 28: Full system integration, with all functionalityimplemented. No additional

features will be added after this date.

April 28-May 4: testing and debugging.

May 4: Public demo. First draft of documentation.

May 4-12: an yadditional testing and debugging, finalization of documentation.

May 12: Final submission.